Author's response to reviews

Title: Different associations between obesity and impaired fasting glucose depending on serum gamma-glutamyltransferase levels within normal range: a cross-sectional study

Authors:

Nam Soo Hong (kusmac25@knu.ac.kr)
Jeong-Gook Kim (mattrix13@naver.com)
Yu-Mi Lee (for2ej2@hanmail.net)
Hyun-Woo Kim (jamesinfm@gmail.com)
sin Kam (kamshin@knu.ac.kr)
Keon-Yeop Kim (pmkky@knu.ac.kr)
Ki-Su Kim (hs.rainmaker@hanmail.net)
Duk-Hee Lee (lee_dh@knu.ac.kr)

Version: 2 Date: 28 March 2014

Author’s response to reviews: see over
Dear Editor in Chief:

We have enclosed our manuscript entitled “Different associations between obesity and impaired fasting glucose depending on serum gamma-glutamyltransferase levels within normal range: a cross-sectional study”. The content of this manuscript has not been published and is not under consideration for publication elsewhere. All authors have participated sufficiently in the conception and design of the work or analysis of the data, as well as the writing of the manuscript. In addition, we reviewed the final version of the manuscript and approved it for publication. We will also take public responsibility for it and believe that the manuscript represents valid work. We will produce the data upon which this manuscript is based for examination by the editors or their assignees if requested.

We are looking forward to hearing a good response.

Sincerely,

DUK-HEE LEE, M.D.
Department of Preventive Medicine,
School of Medicine, Kyungpook University
680 Gukchaebosang-ro, Jung-gu, Daegu, 700-842
Telephone: +82-53-420-6960
Fax: +82-53-425-2447
E-mail: lee_dh@knu.ac.kr